Author’s response to reviews

Title: ARCADO - Adding random case analysis to direct observation in workplace-based formative assessment of general practice registrars

Authors:
Gerard Ingham (drgingham@gmail.com)
Jennifer Fry (jfry@beyondmedical.com.au)
Simon Morgan (simon.morgan@gptvtc.com.au)
Bernadette Ward (bernadette.ward@monash.edu)

Version: 2 Date: 21 Nov 2015

Author’s response to reviews:

We have complied with both requests in the decision letter.

We have added a statement to the methods about informed consent "Written informed consent for participation in the study was obtained from all medical educators and registrars"

We have changed how we reported the authors contributions. Instead of stating that "all researchers were involved in." we have changed to "GI,JF, SM, and BW were involved in...."

These changes have been highlighted in the resubmitted manuscript